Author's response to reviews

Title: Caesarean section rates in Mozambique 1995-2011

Authors:

Qian Long (qian.long@dku.edu.cn)
Taina Kempas (tainakempas@hotmail.com)
Tavares Madede (tmadede@gmail.com)
Reija Klemetti (reija.klemetti@thl.fi)
Elina Hemminki (elina.hemminki@thl.fi)

Version: 2 Date: 27 February 2015

Author's response to reviews:

Dear Editor,

Many thanks for the suggestion.

1. We have included page number in the manuscript and line number in the main text.
2. We have added the acknowledgement in page 16.
3. We have added the authors contributions in page 16 as well.

Thanks!
Qian